Author’s response to reviews

Title: Novel EYA4 variant in Slovak family with late onset autosomal dominant hearing loss: a case report

Authors:

Lukas Varga (varga.lukas@gmail.com)
Daniel Danis (daniel.danis.farmacia@gmail.com)
Martina Skopkova (martina.skopkova@savba.sk)
Ivica Masindova (ivica.masindova@savba.sk)
Zuzana Slobodova (suzanne.slobodova@gmail.com)
Lucia Demesova (luc.demesova@gmail.com)
Milan Profant (profant@fnorl.sk)
Daniela Gasperikova (daniela.gasperikova@savba.sk)

Version: 1 Date: 10 Jan 2019

Author’s response to reviews:

Dear editor,

herewith we would take the liberty to send you the revised manuscript. The manuscript was changed according to your instructions received by e-mail January 4th, 2019.

Kind regards,

Daniela Gasperikova